Author’s response to reviews

Title: Hepatitis B/C in the countries of the EU/EEA: a systematic review of the prevalence among at-risk groups

Authors:

Abby Falla (am.falla@rotterdam.nl)
Sanne Hofstraat (Sanne.Hofstraat@rivm.nl)
Erika Duffell (erika.duffell@ecdc.europa.eu)
Susan Hahné (susan.hahne@rivm.nl)
Lara Tavoschi (Ilara.tavoschi@ecdc.europa.eu)
Irene Veldhuijzen (Irene.veldhuijzen@rivm.nl)

Version: 4 Date: 31 Jan 2018

Author’s response to reviews:

The two editorial changes required have been made. The files names of the figures have also been amended for clarity.